Author's response to reviews

Title: Transition overtime in household latrine use in rural Bangladesh: a longitudinal cohort study

Authors:

Tahera Akter (tahera.a@brac.net)
Nepal C Dey (nepal.cd@brac.net)

Version: 7 Date: 1 June 2014

Author's response to reviews: see over
Notes on reviewer’s report

Title: Transition overtime in household latrine use in rural Bangladesh: a longitudinal cohort study

Reviewer name: Benjamin Arnold

Major compulsory revisions

(1) The manuscript has been edited by the Nextgenediting Global Initiative based in United Kingdom.
(2) The log-binomial regression analysis has been revised according to the reviewer’s comments.
(3) The second sentence has been revised following the lines of reviewer.

Minor essential revisions

(4) In the title, correction has been made by replacing proper word.
(5) Correction was done in conclusion of abstract.
(6) Introduction: The word ‘bacteria’ was replaced by pathogens.
(7) The information was reported to the results section.
(8) The sentence has been deleted following the comments by reviewer.
(9) The sentence has already been deleted following the comments by reviewer.
(10) An article was cited to support chosen analysis method.
(11) Use of robust standard error was explained in statistical analysis section.
(12) The sentence was deleted.
(13) The section “operational definition of a sanitary latrine” was moved to an earlier position of the analysis description.
(14) Necessary changes were made in Table 3 to make it clearer.

Reviewer Name: Abera Kumie

The comment has already been addressed through other reviewer’s comments